Author's response to reviews

Title: Orocervical Foetus in Foetu with prenatal sonographic diagnosis: A case report

Authors:

Kolawole T Braimoh (wolebraimoh@yahoo.com)
Adekunle Y Abdulkadir (akyisau@yahoo.com)
Rabiu O Balogun (ybalogun07@yahoo.com)

Version: 5 Date: 17 October 2008

Author's response to reviews:

Dear Editor,

JMCRR

Sir,

The following changes have been made in the manuscript.

The sentence "Giant cervical foetus in fetu could result in obstructed labour and poor pregnancy outcome if vaginal delivery is attempted" now reads "Giant cervical foetus in fetu is extremely rare. It could result in obstructed labour if vaginal delivery is attempted.

We do agree with the reviewer's opinion that delivering this baby by C/S may not ensure survival in view of its giant nature.